Author's response to reviews

Title: Diabetes, Epidemiological Transition and Health Care Costs in Latin America: Greater Preventive Medicine.

Authors:

Armando Arredondo (armando.arredondo@insp.mx)

Version: 7 Date: 22 July 2014

Author's response to reviews: see over
With respect to manuscript # 5566056941289379, I attach a revised version in response to the last suggestions and comments. To this respect, I would like to make the following observations:

- All comments and suggestions about editing and content were accepted.

- Regarding references requested, the new version includes a new list of all the references required.

- They are examples of diseases including package covering universal coverage in Chile.

- In this version, we have clarified the possible causes of why only 30% of patients are kept under control and as this point is taken up in the proposed approach.

- The new version includes sections that were not in the previous version: declaration of interests, author information, surveys and legend to Figure 1.

Thank you and I await your decision.

Sincerely yours,

Armando Arredondo PhD
Senior Researcher
Center for Health Systems Research
National Institute of Public Health